Author's response to reviews

Title: Quantitative methylation profiling in tumor and matched morphologically normal tissues from breast cancer patients

Authors:
Ilse Van der Auwera (ilse.van.der.auwera@telenet.be)
Cathérine Bovie (catherine.bovie@oncomethylome.com)
Cecilia Svensson (cecilia.svensson@oncomethylome.com)
Xuan B Trinh (xuan.trinh@gza.be)
Ridha Limame (ridha.limame@gza.be)
Peter van Dam (peter.vandam@gza.be)
Steven J Van Laere (steven.vanlaere@gza.be)
Eric A Van Marck (eric.van.marck@uza.be)
Luc Y Dirix (luc.dirix@gza.be)
Peter B Vermeulen (peter.vermeulen2@telenet.be)

Version: 4 Date: 1 March 2010

Author's response to reviews:

We have made the revisions to our manuscript as requested:
- we have deleted the figure from the figure legends as one of the reviewers previously requested to omit this figure
- we have removed the word 'Title' preceding the main heading
- we have removed the keywords
- we have removed the underlying in the text
- we have used initials to indicate the authors in the contribution section
- we have cropped our figures